Aliza Glasbergen-Plas\*, Jenny Doetjes and Lisa L.-S. Cheng

# Linking French *wh*-in-situ and context: a choice function analysis

https://doi.org/10.1515/probus-2025-0003 Received December 5, 2022; accepted July 23, 2025; published online September 12, 2025

**Abstract:** French *wh*-in-situ questions may be infelicitous when a quantifier or focus expression precedes the in-situ *wh*-phrase ('intervention effects'). However, these intervention effects are absent in a certain type of context. This paper offers an account of French *wh*-in-situ questions that explains the role of context. We propose that French *wh*-phrases can be interpreted in-situ by means of a contextually bound choice function. We further hypothesise that, due to an ongoing process of language change, there are two groups of French speakers. Both groups can interpret *wh*-in-situ via the choice function but one group may additionally use covert movement. We show how the choice function account along with this 'Two speaker groups hypothesis' accounts for the effect of context on intervention effects, as well as several other longstanding debates regarding the semantics and prosody of French *wh*-in-situ questions.

**Keywords:** French wh-in-situ; context; choice function; intervention effects; extrastrong presupposition

# 1 Introduction

French has multiple ways to form a *wh*-question, one of which is *wh*-in-situ, exemplified in (1a). This strategy exists alongside several types of *wh*-fronted questions, such as (1b). While the *wh*-phrase is moved to the left edge of the sentence in *wh*-fronted questions, it appears to be left 'in-situ' in *wh*-in-situ questions, at the same position as the corresponding element in a declarative (1c). *Wh*-in-situ questions are

<sup>\*</sup>Corresponding author: Aliza Glasbergen-Plas, Centre for Languages and Academic Skills, Delft University of Technology, Jaffalaan 5, 2628 BX, Delft, The Netherlands, E-mail: a.glasbergen-plas@tudelft.nl. https://orcid.org/0000-0001-5940-2651

Jenny Doetjes and Lisa L.-S. Cheng, Leiden University Centre for Linguistics, Leiden University, Reuvensplaats 3-4, 2311 BE, Leiden, The Netherlands, E-mail: j.doetjes@hum.leidenuniv.nl (J. Doetjes), l.l.cheng@hum.leidenuniv.nl (L.L.-S. Cheng). https://orcid.org/0000-0002-1741-4408 (J. Doetjes). https://orcid.org/0000-0002-0350-5640 (L.L.-S. Cheng)

Open Access. © 2025 the author(s), published by De Gruyter. © IT This work is licensed under the Creative Commons Attribution 4.0 International License.

typically used in colloquial French; see, among many others, Coveney (1989a, b), Adli (2006) and Rowlett (2007).

- (1) a. Elle veut vendre ces bibelots à qui? she wants sell these trinkets to who 'To whom does she want to sell these trinkets?'
  - b. À qui veut-elle vendre ces bibelots?
     To who wants-she sell these trinkets
     "To whom does she want to sell these trinkets?"
  - c. Elle veut vendre ces bibelots à son ami.
    she wants sell these trinkets to her friend
    'She wants to sell these trinkets to her friend.'
    ((1a,b) adapted from Obenauer 1976: 7, (15a, 16a))

The property of French *wh*-in-situ questions that is central to this paper concerns 'intervention effects', exemplified in (2a) versus (2b). The term 'intervention effects' refers to the phenomenon whereby *wh*-in-situ is infelicitous when the *wh*-phrase is preceded by a so-called 'intervener', such as *personne* 'no one' in (2a) (Beck 1996; Bošković 1998, 2000; Chang 1997; Mathieu 1999; cf. Obenauer 1976). *Wh*-fronting, as in (2b), is not blocked by such an expression. The 'interveners', the expressions that induce intervention effects, include quantificational expressions, focus expressions and negation. (Throughout the paper, interveners will be indicated by underlining.)

- (2) a. # <u>Personne</u> n' admire **qui**? no.one NE admires who (adapted from Chang 1997: 20, (40d))
  - b. **Qui** est-ce que <u>personne</u> n' admire? what is-it that no.one <sub>NE</sub> admires 'Who does nobody admire?' (adapted from Chang 1997: 65, (38d))

Intervention effects are commonly assumed to arise when an intervening expression blocks the *wh*-phrase from moving covertly to the left periphery of the sentence, as visualised in (3) (Beck 1996; Bošković 2000; Starke 2001 for French, among others).



<sup>1</sup> The # sign in (2a) indicates that the sentence is grammatical under an echo interpretation.

(3) indicates that the wh-phrase quoi 'what' in (2a) cannot undergo covert movement due to the presence of the intervener personne 'no one' (see Beck 1996; Guerzoni 2006; Kotek 2018, among others). In the grammatical (2b), on the other hand, the wh-phrase moves overtly to Spec CP, showing that overt wh-movement is not sensitive to interveners.

Nevertheless, a French wh-in-situ question containing an intervener may become acceptable when placed in a certain type of context (Baunaz 2005, 2011, 2016; Beyssade 2006; Engdahl 2006; Starke 2001). This is illustrated with the intervener tous les N 'all the N' in (4) as compared to (4'), where the wh-in-situ question is placed under a context.

- (4) # Tous les témoins ont reconnu qui? (without context) all the witnesses have recognized 'Whom did all the witnesses recognize?'
- (4')During a trial, witnesses and defendants are confronted. One of the defendants has been accused unanimously. The journalist asks: Et tous les témoins ont reconnu qui ? (Baunaz 2016: 157, (40b))

Baunaz (2005, 2011, 2016), building on Starke (2001), suggests that this effect of context on intervention effects stems from an interaction between the feature compositions of the intervener and the wh-phrase that crosses it in covert movement. The features represent semantico-pragmatic meanings relating to the kind of context in which a wh-phrase may be used.

Our proposal takes Starke's and Baunaz's insights regarding the type of context that voids intervention effects as a starting point yet takes a different approach. We propose that there exists a mechanism to interpret the wh-phrase in-situ without covert movement; hence, the intervention effects configuration in (3) does not arise. Nonetheless, this mechanism is only available in a certain type of context. Our proposal thus explains the effect of context on intervention effects in a manner that relates it to the absence of (covert) wh-movement. We further hypothesise that covert movement is not available to all speakers of French, while the other mechanism is, leading to what we will call the 'Two speaker groups hypothesis'.

More specifically, we argue that French wh-in-situ questions can be resolved via a contextually licensed choice function (cf. Kratzer 1998 for specific indefinites), which allows a *wh*-phrase to be interpreted in-situ yet imposes restrictions on the context. These restrictions are due to a recoverability condition: the referent selected by the choice function must be recoverable for the interlocutor. The analysis accounts for the absence of intervention effects in certain contexts, because when the choice function is used instead of covert movement, the intervention effects configuration in (3) does not arise. A range of further data are also explained by this account, such as the so-called 'extra-strong presupposition' (Chang 1997; Coveney 1989a, b), and differences between interveners (Baunaz 2011) and among speakers (Adli 2006; Bošković 2000) regarding intervention effects, as well as data variation concerning the presence of a large sentence-final rise in prosody (Adli 2006; Cheng and Rooryck 2000).

The paper is structured as follows. Section 2 presents the generalisation, which motivates the analysis, the 'Intervention effects avoidance generalisation', according to which intervention effects are voided when the referent for the *wh*-phrase has already been established in the context. This section is partly based on re-interpreted insights from Starke (2001) and Baunaz (2005, 2011, 2016). Section 3 introduces the contextually bound choice function, which is subject to a recoverability condition according to which the referent for the wh-phrase must be recoverable in the context. Together, Sections 2 and 3 explain the relation between context and the absence of intervention effects. Section 4 then shows that the choice function analysis also sheds light on (a) the origin of the so-called 'extra-strong presupposition' and that it accounts for (b) differences between interveners and (c) among speakers regarding intervention effects. Section 5 introduces the 'Two speaker groups hypothesis'. It shows how the choice function analysis in combination with this hypothesis accounts for two further areas of data variation: (d) variation regarding the 'extra-strong presupposition' and (e) the presence of a large final rise in prosody. This section also lays out what the 'Two speaker groups hypothesis' predicts regarding intervention effects. Finally, Section 6 compares the proposal to Baunaz (2005, 2011, 2016), which built on Starke (2001). It also discusses some implications of the proposal, including the possibility of an (ongoing) language change.

# 2 Lack of intervention effects in relation to context: a generalisation

As shown by Starke (2001), Engdahl (2006) and in particular Baunaz (2005, 2011, 2016), intervention effects are voided when a wh-in-situ question containing an intervener occurs in a particular type of context. Let us first return to the example in (4'), repeated in (5):

(5) During a trial, witnesses and defendants are confronted. One of the defendants has been accused unanimously. The journalist asks: Et <u>tous les témoins</u> ont reconnu qui ? (Baunaz 2016: 157, (40b))

As Baunaz (2005, 2011, 2016) describes it, the *wh*-phrase *qui* 'who' is felicitous in this context because it is already known that there is a specific defendant who has been accused (i.e. recognized) by all the witnesses. The answer to the question makes

reference to a particular individual that the journalist has in mind. The journalist merely asks for the identity of this already familiar individual. Starke (2001) and Baunaz (2005, 2011, 2016) assume that a wh-phrase that is felicitous in a context as in (5) has the feature 'specific'.<sup>2</sup>

Note, however, that ordinary specific indefinites, that is, non-interrogative specific indefinites, are felicitous without the type of context that is necessary to license wh-in-situ in (5), as is shown in (6).

(6) Oh, I wanted to tell you. Yesterday a certain/this/some man called and asked me a favour.

Thus, the contextual restrictions do not follow from the wh-phrase being treated as a specific indefinite. Rather, if a wh-phrase can felicitously be used in combination with an intervener, it needs to be co-referential with a specific, contextually introduced individual. In the case of (5), this is the defendant that has been unanimously accused.

The relevant property of the context in (5) is thus that it specifies a unique referent for the wh-phrase, namely a defendant who has been recognised by all the witnesses. Any congruent answer to the wh-in-situ question refers to this referent. To see this, suppose that the accused defendant in (5) is defendant number 1, who is called monsieur Bisset, and that he is sitting on the left. In that case, possible answers to the question in (5) could be that all the witnesses recognized defendant number 1, that they all recognized monsieur Bisset, or that they all recognized the defendant on the left. The definite descriptions 'defendant number 1', 'monsieur Bisset' and 'the defendant on the left' would all refer to the same referent, which is the defendant who has been unanimously accused. As a result, the wh-in-situ question in (5) becomes very close to the English paraphrase in (7). We will refer to this type of questions, which intend to identify a previously introduced referent, as 'identity questions' (cf. Rothstein 2001).

(7) The defendant who has been recognized by all the witnesses – who is it? In other words, we suggest that a context that voids intervention effects makes information salient that allows for unique identification of a referent for the whphrase. This means that any answer to the wh-in-situ question refers to the same referent, the one already specified by the context.<sup>3</sup>

<sup>2</sup> See Section 6.1 for more detailed discussion of Baunaz's analysis and how it compares to the analysis proposed here.

<sup>3</sup> A reviewer wondered whether the type of interpretation described here could be analogous to the one found for a fronted wh-phrase without est-ce que (see (1b) above). As argued by Faure and Palasis (2021), these questions have an 'exclusivity reading': at least one referent (i.e. a possible alternative) is excluded as a felicitous answer. Such a reading is nevertheless much weaker than the one found for wh-in-situ questions containing an intervener, where the referent of the wh-phrase is fixed by the context.

In order to understand what it takes for a context to uniquely establish such a referent, we discuss two examples in which the context does *not* establish such a referent and does *not* void intervention effects, the first of which is (8).

(8) Tom is the family globe trotter. He travelled all around the world for more than 20 years. During a family supper, his curious niece presents him a map of the world, with a list of names of all the countries in the world. She asks him:

```
??Tonton
          Tom.
                 ť
                       es
                            pas
                                  allé
                                         où?
 uncle
          Tom
                            not
                                         where
                 vou
                       are
                                  gone
 Intended: 'Uncle Tom, where didn't you go?'
 (Baunaz 2016: 155, (36a))
```

While the context in (5) above makes salient a specific defendant who has been accused by all the witnesses, the context in (8) does not introduce specific places that Uncle Tom did not visit. What the context makes salient is that Uncle Tom visited many places all over the world, but the intervener *pas* 'not' does not feature in the context. Consequently, there is no referent in the preceding discourse corresponding to 'the places where Uncle Tom did not go'. Accordingly, the question in (8) cannot be paraphrased by the identity question in (9).

(9) The places where you did not go – what are they?

This example illustrates that the intervener must be included in what the context makes salient (cf. Beyssade 2006; Engdahl 2006).

A second example in which the context does not identify a unique referent and does not yield intervention effect cancellation is shown in (10).

(10) During the end-of-year party, various prizes were awarded to the best students: maths, English, French, physics, etc. This year, all the students got a prize. After the party, the dean's husband asks his wife:

```
*Tous les étudiants ont reçu quoi / quel prix? all the students have received what which prize Intended: 'What/which prize did all the students receive?' (adapted from Baunaz 2016: 156,(39b))
```

Although the context in (10) mentions the intervener *tous les étudiants* 'all the students', the *wh*-in-situ question is not acceptable on the relevant scope reading. The problem seems to be the scope configuration. The context mentions a set of prizes (maths, English, etc.) and makes salient that all the students got a (different) prize. Nonetheless, there is not a specific prize mentioned in the context such that all the students received it. As a result, the context does not permit the identification of a

unique referent for the wh-phrase, namely a particular prize received by all students. The example in (10) can, therefore, not be paraphrased by the identity guestion in (11).

The prize that all the students received – what/which one is it? (11)

Hence, while a context that voids intervention effects (such as the one provided in (5))<sup>4</sup> allows for the unique identification of a referent for the wh-phrase, a context that yields intervention effects (such as (8) and (10)) does not. Importantly, the context must make salient a specific referent with the property as mentioned in the question, such as the property of having been recognized by all the witnesses in (5). The property crucially includes the intervener (in the right scope configuration). We suggest that this unique identification of the referent in the answer is what characterises contexts that void intervention effects.

To render this generalisation more precise, we employ the notion of entailment by a 'Contextually Salient Meaning (CSM)', used by Büring (2016) to describe the relation between the context and a constituent that is given. Büring's (2016) concept of a Contextually Salient Meaning is illustrated with (12). Mentioning Sinatra in the first part of (12) makes the referent of Sinatra contextually salient; the relevant CSM is thus the singer 'Frank Sinatra'. When the singer is mentioned later in the sentence, its referent is already salient, which makes the singer given. The capitals in (12) represent the main pitch accent of the sentence and  $[...]_G$  represents givenness; in English, givenness is usually associated with deaccentuation (e.g. Selkirk 1984).5

(12)Sinatra's reputation among industry musicians grew swiftly, and *James always supPORted [the singer]* $_{G}$ . (Büring 2016:18, (1))

Following Schwarzschild (1999), Büring uses entailment to regulate which part of a sentence is given in a particular context. A simplified version of this entailment

(Schwarzschild 1999: 145, (11))

The example shows that givenness does not always impose deaccenting. This is relevant for wh-insitu, as we will argue that the *wh*-phrase is given, despite the fact that it is not deaccented.

<sup>4</sup> A further example can be found in Section 6.1, where we compare Baunaz' analysis with ours.

<sup>5</sup> In the context of a narrow focus on a given expression (e.g. a pronoun), the accent is present, as illustrated in (i).

<sup>(</sup>i) A: Who did John's mother praise?

B: She praised [HIM]<sub>F</sub>

relation, which abstracts away from the difference between propositions and other semantic types, is stated in (13).<sup>6</sup>

(13) An expression is given if there is a CSM that entails it.

Contextually salient meanings depend mainly on the context, but Schwarzschild (1999) and Büring (2016), among others, note that such entailment relations may be influenced by world knowledge and associations as well. Consider the example in (14), in which *New Yorker* can be deaccented as given because 'Woody Allen' is made salient by speaker A's utterance.

(14) A: They invited Woody Allen as their keynote speaker.

B: *Yeah*, *they WANted a [New Yorker]*<sub>G</sub>. (Büring 2016: 129, (51); cf. Schwarzschild 1999: 153, (29))

In (14), the relation between *Woody Allen* and *New Yorker* involves the speakers' world knowledge that Woody Allen is a New Yorker. Glasbergen-Plas (2021) suggests that in examples such as (14), the relevant CSM *as it is perceived by the speaker* is not exactly 'Woody Allen', but rather 'Woody Allen', along with the association made salient upon hearing *Woody Allen* that 'Woody Allen is a New Yorker'. This is similar to van Deemter (1994), who states that 'the key is always in the knowledge of the speaker, but what counts is his or her knowledge *about* the reference of certain expressions in the discourse' (p. 25); see Rochemont (1986: 49) for a similar point. We, therefore, use the definition of a CSM in (15), taken from Glasbergen-Plas (2021), which is a modified version of Büring's (2016) CSM, reminiscent of Rochemont (1986: 49). The importance of the speaker's perspective in this definition will become relevant in Section 4.3.

(15) CONTEXTUALLY SALIENT MEANING (CSM)

A meaning is a CSM if it is perceived by the speaker as contextually salient and the speaker has no reason to believe that it is not salient for the addressee.

We are now in a position to reformulate the generalisation regarding the contexts that void intervention effects in terms of entailment. This 'Intervention effects avoidance generalisation' is presented in (16).

<sup>6</sup> Entailment is a relation between two propositions, while in principle every part of a sentence can be given. To ensure that the notion of entailment can be used to define givenness for parts of sentences as well, Schwarzschild, followed by Büring, assumes an 'existential type shifting' operation that turns expressions into propositions. This operation is not relevant for our analysis, as we will only be dealing with propositions.

(16)Intervention effects avoidance generalisation Intervention effects are voided when a Contextually Salient Meaning (CSM) entails the answer to the *wh*-in-situ question.

Let us reconsider the example in (5), repeated as (17) for convenience.

(17)During a trial, witnesses and defendants are confronted. One of the defendants has been accused unanimously. The journalist asks: tous les témoins ont reconnu aui? and all the witnesses have recognized who 'And whom did all the witnesses recognize?' (Baunaz 2016: 157, (40b))

In (17), the relevant CSM is 'a specific defendant has been unanimously accused by all the witnesses'. As discussed above, the referent for the in-situ wh-phrase is this unanimously accused defendant. If this defendant is defendant number 1, called Monsieur Bisset, who is sitting on the left, the answers listed in (18) all refer to this same referent and are congruent answers to the wh-question. Due to this coreferentiality, the CSM entails each of these answers. In (18), phrases with the same index are co-referential.

(18)context: [a specific defendant], has been unanimously accused, i.e., by all the witnesses

#### entails

answer: all the witnesses recognized [monsieur Bisset]; or all the witnesses recognized [defendant number 1]<sub>i</sub> or all the witnesses recognized [the defendant on the left]<sub>i</sub>

As the context involves a CSM that entails the answer to the question, intervention effects are predicted to be absent in (17) according to the descriptive generalisation in (16).

To summarise, this section re-analysed certain examples from the literature regarding the effect of context on intervention effects (Starke 2001; Engdahl 2006; and in particular Baunaz 2005, 2011, 2016). We proposed a more precise generalisation of the relevant contextual property. That is, intervention effects are voided when a Contextually Salient Meaning (CSM) entails the answer to the wh-in-situ question (16). This is only possible if the referent corresponding to the wh-phrase is fixed: The context must make salient a specific referent with the property as mentioned in the wh-in-situ question (such as the property of having been recognized by all the witnesses in (5)).

# 3 Choice function wh-in-situ

We now address the question of why intervention effects can be voided when a CSM entails the answer to the wh-in-situ question. In a nutshell, the explanation we propose is as follows. Recall from Section 1 the standard assumption that intervention effects arise when an intervener blocks covert movement of the wh-phrase, as was visualised in (3), repeated here as (19).

Under this assumption, when a French wh-in-situ question does not display intervention effects, it is logical to assume that covert movement has not taken place. (We leave the discussion of whether covert movement may take place in French in the absence of an intervener to Section 5.) This means that some other mechanism enables the wh-phrase to be interpreted in-situ. The mechanism we propose for this is a contextually bound choice function (cf. Kratzer 1998 for specific indefinites). As the choice function can be used to interpret an in-situ wh-phrase, covert movement does not need to be used, thus avoiding intervention effects. Furthermore, we argue that the contextual binding of the choice function is responsible for the presence of a felicity condition (cf. Ionin 2006) on uttering these wh-in-situ-questions. This condition holds that a choice function is only felicitously used by a speaker when the referent picked out by the choice function is recoverable for the interlocutor. To satisfy this recoverability condition, the referent for the in-situ wh-phrase must be identifiable, which accounts for the contextual restriction found in the previous section. We represent an in-situ wh-phrase that is subject to such a contextual restriction as a choice function f, applying to a particular set. For instance, the in-situ wh-phrase qui 'who' in (17) is represented as f(person), where f is a choice function applying to the set of persons that picks out the defendant who was recognised by all the witnesses. In what follows, we first discuss the interpretation of the wh-phrase, before turning to the semantics of the wh-in-situ question as a whole and the possible answers it may have.

Choice functions are widely used to interpret indefinite noun phrases. A choice function is a function that applies to a non-empty set and yields an individual member of the set (see Reinhart 1998; Winter 1997). Reinhart (1998) extends the use of choice functions to in-situ wh-phrases in multiple wh-questions in English.7 For

<sup>7</sup> For pronominal wh-phrases such as who, we assume that who integrates a choice function variable and a nominal predicate person, resulting in the representation 'f(person)'. As for the moved whphrase which lady in (20), Reinhart (1998) notes that she does not treat it as 'f(lady)' because it has

instance, an in-situ wh-phrase like which book in (20) can be represented by a choice function that applies to the set of books and selects one specific book from the set, as illustrated in (20).

- Which lady e read which book? (20)a.
  - For which  $\langle x, f \rangle$ , (lady(x)) and (x read f(book)) b. (Reinhart 1998: 41, (23a,b))

While the wh-phrase which lady in (20) is interpreted via movement, the wh-phrase of interest here, which book, is interpreted by applying the choice function f to the set of books. The function variable is then unselectively bound by the moved whexpression which. We extend this analysis to single wh-phrases in French wh-in-situ questions, arguing that the lack of a moved wh-phrase is responsible for the lack of a sentential binder for the choice function. This in turn accounts for the special type of interpretation that is found for these wh-in-situ cases, as illustrated in Section 2 above.

We assume that the choice function, without an unselective binder (e.g. introduced by a moved wh-phrase as in (20)), needs to be contextually bound, and that this is an available option in French (see Section 6.2 for cross-linguistic variation in this respect). In this, our approach is similar to that of Kratzer (1998), who assumes that specific indefinite determiners such as a certain and some introduce choice functions that are free variables whose value is determined by the context.8

As argued by Breheny (2003) and Yanovich (2005), contextually assigned values are typically recoverable for the interlocutor. If the interlocutor cannot identify this value, the communication fails. As we indicated above, a choice function in a wh-insitu question is recoverable from the context, if the context makes it possible for the interlocutor to identify the intended referent of the wh-phrase. In (17), the context introduces a specific referent corresponding to the 'person whom all the witnesses recognized'. Under our analysis, the choice function f in this wh-in-situ question yields this particular referent when applied to the set of persons. However, when the referent picked out by the choice function is not recoverable from the context, the

moved to SpecCP. She nevertheless acknowledges that the wh-phrase could be reconstructed at LF, which would make it possible to interpret the reconstructed wh-phrase in the same way as the in-situ wh-phrase which book, corresponding to f(book) in (20b). Following Reinhart, we will leave open the question of whether the moved wh-phrase is interpreted by means of a choice function as well, or whether this analysis only applies to wh-phrases that are left in-situ.

<sup>8</sup> Kratzer's (1998) contextually supplied choice function has been criticised by Breheny (2003) and Yanovich (2005), but their criticisms mainly concern the implicit argument that Kratzer uses to take care of specific indefinites like a certain. Our proposal is different, as it does not employ the implicit argument.

interlocutor cannot identify the referent intended by the speaker. This makes the whin-situ question containing the choice function infelicitous (cf. Ionin 2006).

We now turn to the proposed semantics of choice function wh-in-situ. In order to do so, we first briefly discuss the interpretation of wh-questions containing a moved wh-phrase, which we assume to denote a set of alternative propositions (Hamblin 1973). We adopt the idea that the shift from a proposition to a set of propositions is made by means of an operator (Karttunen 1977), which we locate in C following Heim (2000) and Dayal (2016). This yields a proto-question (see Karttunen 1977) consisting of a set containing a proposition with a free variable in the position of the whphrase's trace. Consider the proto-question exemplified in (21a) with the question operator (indicated by ?) and its complement  $[I_P t_i walks]$ . When the question operator (defined in (21b)) is applied to its complement (which contains the free variable  $t_i$ ), it derives (21c), a set containing that proposition (cf. Dayal 2016: 27, (6)). (21d) corresponds to the set notation of (21c).<sup>9</sup>

```
(21)
       b. [?] = \lambda p1 \lambda p2 [p2 = p1]
              [?]([t_i \text{ walks}]) = \lambda p1 \lambda p2 [p2 = p1] (\lambda w [walks(t_i)])
        c.
                    \Rightarrow \lambda p [p = \lambda w [walks(t_i)]]
              \{p \mid p = t_i \text{ walks in } w\}
        d.
```

To form the complete question which student walks, the wh-phrase, which student, is added, which functions as the binder for the free variable.

The outcome of combining the proto-question with the wh-phrase needs to be a set of propositions such that for each proposition there is a student that walks, as given in (22) (both as a lambda expression and in set notation):

```
(22)
                       \lambda \mathbf{p} \exists \mathbf{x} [student(\mathbf{x}) \& \mathbf{p} = \lambda \mathbf{w} [walks(\mathbf{x})]]
                        \{p \mid \exists x \mid student(x) \& p = x \text{ walks in } w\}
```

To arrive at this semantics, the existential quantifier that is introduced by the wh-phrase in SpecCP needs to quantify into the proto-question (Karttunen 1977), as illustrated in (22a). As a result, the existential quantifier will be located between the lambda operator  $\lambda p$  and the part of the formula that identifies p ( $p = \lambda w$  [walks(x)]),

<sup>9</sup> We adopt a modified version of the question operator, following Heim (2000). Whereas Hamblin defines questions as sets of propositions that count as possible answers to the question, Karttunen assumes that the set of propositions only contains true propositions. This difference is not essential to our approach. We assume that ordinary wh-questions correspond to sets of possible answers, because this makes it conceptually easier to see the difference between ordinary wh-questions and wh-in-situ questions containing a contextually bound choice function. The variable w in (21c) is a possible world variable, which is interpreted as the actual world in (21d).

rather than in the position outside the scope of the lambda operator where the whphrase is generated.

Quantifying-in can be captured either by the semantics of the *wh*-phrase or by a general rule. Dayal (2016) shows that the latter option permits the moved wh-phrase to be interpreted as a generalised quantifier, on a par with other indefinite determiners (see (23a)). Her version of the rule is given in (23b); see (21) above for an illustration of the semantics of the proto-question [C'].

The quantifying-in rule in (23b) does two different things. On the one hand, it turns the proto-question [C] into a predicate over possible referents of the trace by adding a lambda operator  $\lambda x_n$ , which binds the trace of the wh-phrase. This provides the generalised quantifier which N with an argument of type  $\langle e,t \rangle$ . On the other hand, the rule relocates the lambda expression inside [C'] to a position above the *wh*-phrase introducing the existential quantifier.<sup>10</sup> As a consequence, the proto-question, a singleton set, is transformed into a meaningful question: a set of alternative propositions that, from the perspective of the speaker, could be either true or false.

We are now in a position to consider the interpretation of in-situ questions containing a choice function. Taking again (17) as an example, the question operator ?, as repeated in (24a), turns the proposition in the IP in (24b) into the set in (24c); (24d) defines the same set in set notation:

- (24) $[?] = \lambda p1 \lambda p2 [p2 = p1]$ 
  - [CP ? [IP all the witnesses have recognized f(person)]]
  - $\lambda p1 \lambda p2 [p2 = p1] ( [[p2 = p1] ( [p2 =$ c.  $\Rightarrow \lambda p [p = \lambda w \ \forall x [witness(x) \rightarrow has recognized (x, f(person))]]$
  - d. {p | p = all the witnesses have recognized f(person)}

Given that the contextual binder of the choice function f in this example is not located in the CP, the quantifying-in rule in (23b) cannot apply. As this operation is crucial for turning the singleton set into a set of alternative propositions, this implies that the set that results from the application of the question operator in (24) remains a protoquestion, that is, a singleton set. 11

<sup>10</sup> To be precise: it resolves the original variable inside C' by a new variable (p) and adds a binder  $(\lambda p)$ for this new variable *p* to the left of the *wh*-phrase.

<sup>11</sup> The impossibility of applying the quantifying-in operation to a binder that is external to the sentence excludes the possibility of creating a set of alternative propositions in which the choice function in each proposition is bound by a different contextual binder.

This raises the question of how the resulting representation (24c), corresponding to the wh-in-situ question in (17), can be interpreted. Being a singleton set, it 'does not represent non-trivial choice between two or more propositions' (Heim 2000: 8, interpreting Karttunen 1977). However, the realisation of the unique proposition introduced by a question interpreted via a contextually bound choice function is far from trivial, at least from the perspective of the speaker. Imagine that the speaker of (17) and the interlocutor are both watching the defendants, who are respectively sitting to the left, in the middle and to the right, and that both the speaker and the interlocutor know that one specific defendant has been recognized by all the witnesses, but the speaker does not know which one. By uttering the wh-in-situ question in (17), the speaker indicates that s/he does not know which of the imaginable realisations of the proposition in (25) matches the actual world and thus invites the interlocutor to realise the proposition in a more informative way. (The index in (25) is added to express co-reference with the referent introduced by f(person).)

- (25)Imaginable realisations of the single proposition introduced by the question in (17)/(24c) from the perspective of the speaker
  - All the witnesses recognized [the defendant to the left]<sub>i</sub>.
  - b. All the witnesses recognized [the defendant in the middle]<sub>i</sub>.
  - All the witnesses recognized [the defendant to the right]<sub>i</sub>.

Hence, if (25a) is the answer to the question in (17), by uttering (25a), the interlocutor provides the information to the person who uttered the *wh*-in-situ question in (17) that the contextually introduced defendant who was recognized by all the witnesses is the one to the left. This is what gives these wh-in-situ questions the flavour of identity questions of the type in (7), (9) and (11). Although the set in (24c) contains no alternative propositions, the question in (17) still elicits information that was hitherto not available to the speaker. We, therefore, suggest that the existence of non-trivially different imaginable realisations of the proposition (from the perspective of the speaker) turns the question in (17)/(24c) into a non-trivial one, despite the fact that it introduces a single proposition.<sup>12</sup>

<sup>12</sup> Alternatively, one might argue that wh-in-situ questions that are interpreted by means of contextually bound choice functions are not real questions, but declaratives (cf. Biezma 2018). According to Biezma, these declaratives receive question force, because they function as non-informative answers to the question under discussion that they evoke. Being non-informative, they will not answer the QUD, which thus remains open and can be answered by the interlocutor. Within our proposal, these declaratives would fix the reference of the expression that corresponds to the whphrase, thus accounting for their context sensitivity. Note that the contextual restriction formulated by Biezma (for English cases of wh-in-situ) is much weaker than the one laid out in Section 2, which was motivated by the contextual restrictions found in *wh*-in-situ questions containing interveners.

Summarising this section, we adopt Kratzer's (1998) contextually supplied choice function and apply it to wh-in-situ questions. The wh-phrase is interpreted in-situ by means of a choice function, which is contextually bound. As there is no covert movement, intervention effects do not arise even when an intervener is present. We suggest that the choice function is only available when the context establishes a unique referent for the wh-phrase, because a felicity condition on uttering these whin-situ questions states that a choice function is only felicitously used by a speaker when it is recoverable for the interlocutor. Adopting a semantics combining insights from Hamblin (1973) and Karttunen (1977), we argued that the contextual binder, other than a moved wh-phrase, cannot undergo the operation that is necessary to turn the question into a set of alternative propositions. As a consequence, these questions denote sets containing a single proposition and a congruent answer to them involves providing more information about a previously established referent, corresponding to the wh-phrase. Thus, the semantics of a wh-in-situ question containing a choice function in (24) ties together the lack of (covert) wh-movement and the fixed referent for the *wh*-phrase.

# 4 Further facts explained by the choice function analysis

Sections 2 and 3 together explained why certain contexts void intervention effects: the choice function, which is only licensed when its value is recoverable for the interlocutor, interprets in-situ wh-phrases without covert movement. The current section describes how the choice function analysis also accounts for a range of further data. The analysis predicts a phenomenon that has been referred to in the literature as the 'extra-strong presupposition' (Section 4.1). Regarding the compatibility of wh-in-situ with expressions that can provoke intervention effects, it explains the observed differences between interveners (Section 4.2) and among speakers (Section 4.3). In the remainder of the paper, we refer to the questions interpreted via the choice function mechanism as 'choice function wh-in-situ'.

# 4.1 The extra-strong presupposition

The so-called 'extra-strong presupposition' is usually attributed to Chang (1997), although she cites Coveney (1989a). It is illustrated by (26) and (27), which display a wh-fronted and a wh-in-situ question, each accompanied by an answer.

(26)Ouestion: *Ou*' est-ce que Marie a acheté? Answer: Rien. what is-it that Marie has bought nothing 'What did Marie buy?' 'Nothing.'

Answer: ?? Rien (27)Ouestion: Marie a acheté quoi? Marie has bought what nothing 'What is it that Marie bought?' 'Nothing.' (adapted from Chang 1997: 42, (37), (40))

The wh-fronted question in (26) is a neutral question. Like other wh-questions, it involves an existential presupposition or implicature: the speaker expects there to be an answer to the question. The question in (26) can potentially receive a negative answer like rien 'nothing' (although this answer would not be expected). In contrast, the wh-in-situ question in (27) has been claimed to be felicitous only if the speaker already assumes that Marie bought something, since it is 'strongly presupposed' that there exists a value to fill the wh-phrase. The speaker merely requests more detail about the purchase, as in: what it is that Marie bought. Therefore, a negative reply like *rien* 'nothing' is quite odd.

We suggest that the questions that were claimed to exhibit an extra-strongpresupposition involve choice function wh-in-situ. To license (27) as choice function wh-in-situ, that is, for the value of the choice function to be recoverable, the referent for the wh-phrase must be identifiable in the preceding context: the entity that Marie bought must already be salient for the speaker. The speaker merely requests further information about this entity. The reply that Marie bought nothing would then not only be an unexpected answer but also an incongruent one.

Another example that has been claimed to exhibit an extra-strong presupposition is given in (28). This example illustrates the relevance of the speaker's associations based on world knowledge for this alleged stronger presupposition.

- (28)A: C' est l' anniversaire de Pierre la semaine prochaine. the birthday of Pierre the week it next 'It's Pierre's birthday next week.'
  - vas lui acheter quoi? and you will for.him what buy 'And what will you buy for him?' (Cheng and Rooryck 2000: 24, ftn. 3, adapted from Chang 1997)

Speaker A mentions that Pierre's birthday is coming up. Based on speaker B's knowledge of the world, this leads him/her to assume that there is a particular present that speaker A is planning to buy for Pierre. The question of what this present is, is merely 'seeking details on an already established (or presupposed) situation' (Chang 1997: 46). Under the current proposal, the context in (28) (in combination with world knowledge) can make the referent for the in-situ wh-phrase, the present that speaker A is planning to buy for Pierre, salient for speaker B, which makes the value of the choice function recoverable. To be precise, the relevant CSM is 'semi-salient' in this example (see also Section 4.3): only one possible interpretation of the context makes a specific referent for the wh-phrase salient, while an interpretation that does not license the choice function is also possible. On the interpretation of the context that makes salient a specific referent, the salient present is the unique established referent for the wh-phrase. Speaker B in (28) does not know to what type of present this referent corresponds and thus ignores which imaginable realisation of the single proposition is a true proposition in the actual world. By uttering (28), speaker B requests information about the identity of this previously established referent.

It was previously not clear what the conceptual status was of the extra-strong presupposition or where it might come from. Starke (2001) and Baunaz (2005, 2011, 2016) attribute the alleged presupposition to a particular feature that is part of the internal structure of a strongly presupposed in-situ wh-phrase, but not of a wh-fronted one. In Baunaz's (2011, 2016) terminology, an in-situ wh-phrase with the feature 'specific' or 'partitive' carries a stronger presupposition than a 'non-presuppositional' whphrase. Likewise, Boeckx attributes the stronger presupposition to a definite presuppositional D feature on an in-situ wh-phrase (Boeckx 1999; Boeckx et al. 2001). These proposals place the alleged presuppositional nature of wh-in-situ, which at first sight seems like a semantico-pragmatic phenomenon, in the morphology (nano-syntax) of the wh-phrase. Our proposal is similar in the sense that it attributes a specific or definite-like meaning to the wh-phrase. However, our analysis relates this meaning to the lack of wh-movement and attributes it to a felicity condition. This ties the extrastrong presupposition, the fact that it is associated with in-situ wh-questions in particular, and the lack of intervention effects in such questions all together.

Mathieu (2002) describes the stronger presupposition in terms of background in contrast to focus. Chang (1997), on the other hand, mentions that 'all information other than the questioned element is taken for granted' and that 'the information expressed by everything, except the wh-word is already a salient part of the previous discourse' (p. 44). This intuition is made specific by the current proposal: the referent for the wh-phrase must be established in the context to make the choice function recoverable, predicting the 'extra-strong presupposition'.

Recall from Section 3 that we assume a felicity condition on the use of a choice function (cf. Ionin 2006), according to which the choice function must be recoverable for the interlocutor. Ionin argues for the specific indefinite this (as in There is this book about questions you really should read) that it is subject to a felicity condition, in contrast to a definite article, which introduces a presupposition. Contrary to presuppositions, felicity conditions do not require maximisation. If one can use a definite article rather than an indefinite one, one will do so due to Maximise

Presupposition (Heim 2008). However, specific indefinite this N can be replaced by an N without provoking a problem. Something similar can be observed in wh-in-situ questions. It does not seem to be the case that a context that permits the use of a choice function wh-in-situ question is incompatible with ordinary wh-questions. That is expected if it is a felicity condition that provokes the context dependency. In that sense, the phenomenon referred to as an 'extra strong presupposition' is a combination of the same existential presupposition that is commonly assumed for any wh-question and a felicity condition on the interpretation of the wh-phrase (via the contextually bound choice function).

# 4.2 Choice functions and apparent interveners

We argued in Section 3 that a choice function wh-in-situ question is unacceptable if the referent for the wh-phrase is not retrievable from the context, and that these sentences are not sensitive to intervention effects. On the basis of this, one might expect classical interveners to not induce infelicity; as long as the referent picked out by the choice function is recoverable by the context, the sentence should be felicitous. Nevertheless, the matter is more complicated. We show below that the class of interveners is not homogeneous; some interveners seem to make in-situ wh-questions unacceptable while others appear to be more permissive. In this section, we argue that differences within the class of interveners are due to differences in how difficult it is for speakers to think of a context that satisfies the recoverability condition associated with the choice function. Given this explanation, the unacceptability provoked by the presence of an intervener in choice function wh-in-situ is thus not an intervention effect (see (3) above), and the differences between (apparent) interveners follow from our analysis.

Certain expressions, such as focus adverbs (seulement 'only') and negative expressions ((ne) jamais 'never' and (ne) personne 'no one'), are commonly assumed to result in unacceptability (e.g. Baunaz 2011; Engdahl 2006; Hamlaoui 2010; Mathieu 1999).

(29)\*Seulement Jean arrive à faire quoi? Only Iean arrives to do what Intended: 'What does only JEAN manage to do?' (Mathieu 1999: 447, (12a))

Other expressions are more controversial in that previous literature does not consistently treat them as interveners. This is particularly true of (ne) pas 'not' and several universal quantifiers (toujours 'always', tous les N 'all the N', tout le monde 'everybody' and floating tous 'all'). The examples in (30) and (31) display contradictory judgements of sentences with the same intervener.

- (30)\*Tu ne fais pas quoi not what this evening do vou ne Intended: 'What aren't you doing tonight?' (Mathieu 2002: 35, (2a))
  - b. 11 (ne) doit pas toucher qui? he NF. who must not touch 'Who mustn't he touch?' (Adli 2006: 177, (9a))
- (31)\*Tout le monde a vu auoi? all the world has seen what Intended: 'What did everyone see?' (Mathieu 1999: 464, (45b))
  - Et à cette fête, à ton avis, embrassé tout le monde and at this party in your opinion all the world has kissed qui? whom 'And at this party, in your opinion, whom did everybody kiss?' (adapted from Poletto and Pollock 2015: 86, (20a))

The availability of a contextually bound choice function sheds light on these controversies. First of all, if a sentence is presented to an informant in isolation (i.e. without context), the informant is free to construe their own context, which may or may not license the use of a contextually bound choice function. Crucially, for some interveners, it is easier to envisage a context that makes the choice function recoverable than for others. An example of an intervener where this is relatively easy is *plusieurs* N 'several N'. The sentence in (32) with this intervener was judged acceptable (Adli 2006).13

(32)Plusieurs chênes été coupé où? ont several oaks have been cut where 'Where have several oaks been cut?' (Adli 2006: 180, (16a))

To license a choice function in (32), the context must make salient that there is a specific place where several oak trees have been felled. The speaker then inquires where this place is. The noun phrase plusieurs chênes 'several oaks' can be either

<sup>13</sup> Adli mentions that the sentence was presented with context but does not report on the context itself.

interpreted collectively or distributively. If it is interpreted collectively, the oaks will be all felled in a single place. In this case, the interlocutor may quite easily imagine a context in which there is a salient, contextually given place where these trees have been felled.

In contrast to plusieurs N 'several N', it is more difficult to construe a context that establishes a referent for a question containing (ne) que 'only', as illustrated by (33):

(33)#Iean (n') a parlé qu' à Suzanne dans quelles circonstances? Jean NE has spoken only to Suzanne in what circumstances 'In what circumstances did Jean only speak to Suzanne?'

We suggest that (33) is judged to be unacceptable when presented to informants in isolation because it is difficult to envisage the necessary context, as indicated by the #. 14 For the choice function to be licensed in (33), the context must make salient that there exist particular circumstances under which Jean only spoke to Suzanne. It is not straightforward to envisage a context that makes this salient, as the restriction on the context is guite specific. Nonetheless, a question with the focus expression (ne) que 'only' can be made felicitous given an appropriate context; native speakers we consulted found the example in (34) acceptable. 15 (The intended reading of (34) is that at the events at the end of the year, they did not speak to other people than Suzanne.)

(34)Pierre, Paul et Jean sont tous allés au lycée ensemble. À la fin de l'année, chacun d'eux n'a parlé qu'à Suzanne. Pierre l'a croisée pendant les examens, Paul lui a parlé lors de la remise des diplômes. Pierre, Paul and Jean all went to secondary school together. At the end of the school year, each of them only spoke to Suzanne. Pierre met her during the exam period, Paul spoke to her at the graduation ceremony.'

<sup>14</sup> Many authors mark sentences containing 'apparent interveners' by a star rather than a hash. In the examples that we cite from the literature, we use the judgements as indicated by the authors. 15 We thank Romane Pedro and Marie Pedro for helping us construct the relevant examples and contacting other speakers, who confirmed their judgements. One of our reviewers pointed out that they did not accept the sentence in the presence of ne, because this would involve a clash of registers: more formal due to the presence of ne and informal because of the use of wh-in-situ. We doublechecked the examples with and without ne with several speakers, who accepted them in the relevant context, independently of the presence or absence of ne. Following Adli (2006), who also tested wh-insitu questions with and without ne, we put ne between brackets in (33) and (34). In examples cited from the literature, we follow the authors. Note furthermore that some speakers we consulted would also accept (34) with seulement instead of (ne) que, while others only accepted (34) with (ne) que. We leave the compatibility of ne and wh-in-situ as well as the question why seulement seems less acceptable in *wh*-in-situ, even when the right context is provided, for future research.

Et Jean, il (n') a parlé qu' à Suzanne dans quelles circonstances? and Jean he NE has spoken only to Suzanne in what circumstances 'And Jean, in what circumstances did he only speak to Suzanne?'

Note that the context in (34) is much less common in the sense that it describes a very particular situation (three people only talking to a certain person and not to others in different circumstances) and a communicative setting that is not completely straightforward either. This illustrates the contrast with (32), for which both the situation (a collection of trees being felled) and the communicative context (a discourse referent corresponding to the place where it happened is salient in the context) are much more straightforward. In sum, the differences between apparent interveners can largely be explained by how easily one can construe a context that licenses the use of a choice function for a sentence containing that apparent intervener.

# 4.3 Choice functions and differences among speakers

There is also variation in judgements of sentences containing the same 'intervener', which seems to be due to differences between speakers. <sup>16</sup> An example in which the same sentence receives contradictory judgements is given in (35) (see also (30) and (31) above).

(35)(#) Il (n') rencontré qui? а pas he has who NF. not met 'Whom didn't he meet?' (✓Adli 2004: 203, (3a)/#Chang 1997: 63, (34a))

The choice function mechanism can also shed light on such variation. As was discussed above, informants can always construe (more details of the) context themselves. One informant may envisage another context or a more elaborate one than another. Some informants may be very good at quickly construing the relevant context, allowing them to accept questions with interveners more easily. Recall also from Section 2 that a speaker's associations may influence what a sentence makes salient for them. For example, a person's beliefs about and experiences with children may influence what the question in (36) makes salient.

<sup>16</sup> As was alluded to in the discussion of examples (33)/(34), aspects of the sentence meaning other than the intervener may also affect how easily one can envisage the necessary context, which should be considered when comparing judgements of different sentences (containing the same intervener) (Glasbergen-Plas 2021).

(36) ?Et ta fille, elle ne mange pas quoi ?
and your daughter she NE eats not what
'And your daughter, what doesn't she eat?'
(adapted from Engdahl 2006: 100, (23))

Even if a sentence is presented to an informant in a particular context, this context may allow for more than one interpretation, depending on the speaker. An example is the context in (37), adapted from Engdahl (2006).

### (37) Semi-salient

Anne has two children, a son and a daughter. They are both rather picky about what they eat. Anne mentions that her son doesn't eat fish. Her friend asks:

(?) Et ta fille, elle ne mange <u>pas</u> quoi? and your daughter she NE eats not what 'And your daughter, what doesn't she eat?' (adapted from Engdahl 2006: 100, (23))

Does the remark that both children are *rather picky* establish that there is a specific thing that Anne's daughter does not eat (at all)? That seems to be open to interpretation. By analogy with the son, there may also be a specific type of food that the daughter does not eat. Nevertheless, the daughter could also just be generally fussy about her food. We call the meaning that would not so easily license the use of a choice function in (37) 'semi-salient': the context allows for both an interpretation that makes the choice function recoverable and an alternative interpretation. One can render the context in (37) more explicit so that it leaves no more room for an alternative interpretation (38).

### (38) Salient

Anne has three children, two sons and a daughter. She has a rule at home according to which each child is allowed to have one type of vegetable that they do not eat. Anne mentions that her oldest son doesn't eat cabbage and her youngest son doesn't eat sprouts. Her friend asks:

Et ta fille, elle ne mange <u>pas</u> quoi? and your daughter she NE eats not what 'And your daughter, what doesn't she eat?'

In (38), we (a) added a child, strengthening the sense of analogy, (b) made explicit that each child has the relevant property of not eating a specific thing and (c) made the contextual restriction of the *wh*-phrase more specific (vegetables instead of food items in general). All these elements increase the salience of a specific type of vegetable that Anne's daughter does not eat.

In sum, the choice function proposal predicts the observed variation in judgements. Variation is expected because individual speakers may envisage and interpret contexts differently.

# 5 Two groups of French speakers

subset of the speakers of French.

We have shown in the previous sections how the availability of a contextually licensed choice function ties together the absence of intervention effects in certain contexts and the 'extra-strong presupposition', and how it explains differences between apparent interveners and speakers regarding apparent intervention effects. Nevertheless, as we lay out below, there is strong evidence that French wh-in-situ questions do not give the appearance of an extra-strong presupposition for all speakers. This can be explained if there is also a second mechanism to interpret French wh-in-situ, namely covert movement, which does not impose restrictions on the context. We hypothesise that this second mechanism is only available to a subset of French speakers (39) for interpretating single wh-in-situ. 17,18

(39)TWO SPEAKER GROUPS HYPOTHESIS While the contextually licensed choice function is available to all speakers, covert movement is only available for the interpretation of wh-in-situ to a

We call speakers who only have the choice function option 'Type A speakers' and speakers who have both mechanisms to interpret wh-in-situ 'Type B speakers'.

The idea that the contextual restrictions on wh-in-situ are not the same for all speakers was recently discussed by Baunaz and Bonan (2023) (see also Larrivée 2019; Glasbergen-Plas 2021). Baunaz and Bonan distinguish two varieties of European French: Standard Colloquial (SC) and Non-Standard Colloquial French (NSC). This is similar to Zribi-Hertz' (1994) distinction between Colloquial French and Very Advanced French, both of which differ from Modern Standard French. Modern Standard French is defined as the productive variant of formal French, while both colloquial French and Very Advanced French are both varieties of informal French. Baunaz and Bonan (2023) show on the basis of corpus data from different time periods that the contextual restrictions on wh-in-situ diminish over time, and argue

<sup>17</sup> It remains an open question whether there are also French speakers without a contextually licensed choice function in their grammar for interpreting wh-in-situ. These speakers must use covert wh-movement for all wh-in-situ interpretation. Therefore, they would be predicted to never allow interveners in wh-in-situ questions, irrespectively of the context.

<sup>18</sup> Covert movement is, however, expected to be available to all speakers of French for the interpretation of multiple questions.

on the basis of this that contextually restricted wh-in-situ is a feature of Standard Colloquial French, while wh-in-situ is not contextually restricted in Non-Standard Colloquial French.

The current proposal is in line with the observations of Baunaz and Bonan. However, rather than focusing on the changes in contextual restrictions on wh-insitu over time, we aim at analysing the properties of the two groups of speakers based on our hypothesis in (39). Whereas speakers of group A are restricted to Standard Colloquial French, speakers of group B also draw on Non-Standard Colloquial French (but see footnote 17 above).

The remainder of this section shows how the combination of the 'Two speaker groups hypothesis' and the choice function analysis explains a further set of data: the differences among speakers regarding the extra-strong presupposition (Section 5.1) and regarding the presence of a large sentence-final rise in prosody (Section 5.2). The third subsection discusses the predictions regarding intervention effects for the two groups of speakers (Section 5.3). The section ends with a summary (Section 5.4). Below, we refer to questions interpreted via covert movement as 'covert movement wh-in-situ'.

# 5.1 Differences among speakers: the extra-strong presupposition

The view presented in Section 4.1, according to which French wh-in-situ questions involve an extra-strong presupposition, is quite wide-spread in the literature (Boeckx 1999; Boeckx et al. 2001; Boucher 2010; Cheng and Rooryck 2000; Mathieu 2002; Zubizarreta 2003). However, many other authors maintain that French wh-in-situ questions do not involve a stronger presupposition than wh-fronted questions (Adli 2006; Aoun et al. 1981; Beyssade 2006; Hamlaoui 2011; Mathieu 2004; Oiry 2011; Shlonsky 2012; Starke 2001). A third position holds that only some wh-in-situ questions involve an extra-strong presupposition (Baunaz 2005, 2011, 2016; Coveney 1989a,b; Starke 2001; Zimmermann and Onea 2011; in a different framework Myers 2007).

While our proposal aligns with the last position, the 'Two speaker groups hypothesis' put forward in this section also accounts for the data variation. Recall that the grammar of Type A speakers only has a contextually licensed choice function to interpret wh-in-situ. Thus, for Type A speakers, French wh-in-situ questions require a particular type of context, which makes the referent for the wh-phrase recoverable (e.g. Chang 1997; Cheng and Rooryck 2000). On the other hand, for Type B speakers, who also have the covert movement option, French wh-in-situ questions can also be used out-of-the-blue (e.g. Adli 2006; Hamlaoui 2011).

Direct evidence for the existence of a group of speakers who can interpret wh-insitu via covert movement is to be found in Glasbergen-Plas (2021). The author reports on a rating study investigating the felicity of French wh-in-situ long-distance questions in an out-of-the-blue context (as well as indirect wh-in-situ questions and wh-insitu questions containing an adjunct island). 19 The contexts used in the study were specifically designed to exclude the possibility of licensing via a contextually bound choice function. An example is presented in (40).

(40)Tu es assis dans le bus qui va jusqu'à l'université. Tu parles avec un ami qui suit les mêmes cours que toi. Il te parle de ses projets pour l'été. Soudain, tu lui dis:

> You're sitting on the bus to the university. You're talking to a friend, who is in the same program as you. He tells you about his plans for the summer. Suddenly you say:'

Sinon, ie pense ça, by the way I think of that 'By the way, I'm just wondering,

Tu crois que le prof а prévu quoi pour l' examen? you believe that the teacher has planned what for the exam what do you think the teacher planned for the exam?'

The questions in contexts were rated by monolingual native speakers of French (N =64, 20–35 years of age) who originated from different regions in France. Questions as in (40) received a mean rating of 4.0 on a scale from 1.0 to 5.0. That is, speakers in a relatively young age group accepted French wh-in-situ questions in a context designed to exclude the use of a choice function. Under the current analysis, this confirms the existence of Type B speakers, speakers who have a second mechanism to interpret French wh-in-situ questions in addition to a choice function. The rating study also suggests that the group of Type B speakers includes speakers in a younger age group (age 20 to 35).20

As indicated above, these conclusions are in line with the findings of Baunaz and Bonan (2023), who show that the proportion of wh-in-situ questions of which the propositional contents is not contextually given increases over time.

<sup>19</sup> There is discussion in the literature about the acceptability of long-distance wh-in-situ questions (e.g. Bošković 1998, 2000; Oiry 2011).

<sup>20</sup> Baunaz (2005) suggests that embedded wh-in-situ is only possible with a 'specific' wh-phrase, which is not in accordance with the results of the experiment reported here: both mechanisms seem to allow for embedded wh-in-situ. Note that there is nothing that would block an analysis of embedded wh-in-situ by means of a choice function, provided that the context is sufficiently rich to identify the intended reference of the wh-phrase.

# 5.2 Differences among speakers: the large sentence-final rise

A second area of speaker variation that the 'Two speaker groups hypothesis' sheds light on concerns judgements regarding the presence of a large sentence-final rise (i.e. a rise with a large pitch excursion). We first summarise the debate in the literature on this topic before explaining the current proposal's contribution.

The main significance of the alleged final rise stems from a theoretical paper by Cheng and Rooryck (2000), who propose that wh-in-situ questions are licensed by their rising intonation. A production study by Déprez et al. (2013) provides 'nuanced support' (p. 15) for Cheng and Rooryck's claim, although Tual (2017) could not replicate Déprez et al.'s findings. An opposing view states, however, that French wh-in-situ questions end in a fall (Di Cristo 1998; Mathieu 2002; Starke 2001). Yet a third position holds that a large sentence-final rise is possible, but optional rather than mandatory (Adli 2004, 2006; Wunderli 1978, 1982, 1983; Wunderli and Braselmann 1980). Reinhardt (2019) settles this debate to some extent. In two corpus studies, she shows that while both fronted and in-situ wh-questions occur with a rising as well as a falling final contour, a large sentence-final rise is more frequent in *wh*-in-situ than in *wh*-fronted questions.

A production study by Glasbergen-Plas et al. (2021) throws new light on the observed variation. They investigated the prosody of French wh-in-situ questions with either a broad focus or a narrow focus on the wh-word (as well as echo questions). Although it is sometimes assumed that the focus in wh-questions necessarily equals the wh-phrase, this is not the case in all languages (Beyssade 2006; Eckardt 2007; Jacobs 1984, 1991; Ladd 2009; Reich 2002). Wh-questions in French may be prosodically marked for focus (Beyssade 2006; Glasbergen-Plas et al. 2021). Crucially, Glasbergen-Plas et al. show that French wh-in-situ questions with a narrow focus on the wh-word display a large sentence-final rise, while wh-in-situ questions with broad focus do not.<sup>21</sup> Based on known correlates of focus marking in French, the authors analysed the large sentence-final rise as a correlate of narrow focus marking.<sup>22</sup> In other words, they show that the large final rise observed in some French wh-in-situ questions is a marker of narrow focus on the wh-word.

A key point to note is that choice function wh-in-situ questions always have a narrow focus on the wh-word. As an illustration of this point, consider a context in

<sup>21</sup> Broad focus questions exhibited a mostly falling contour which ended in a very small rise.

<sup>22</sup> In the experiment, the sentence-final rise of the narrow focus questions was a result of 'tone copying' (Martin 1981), the copying of the F0 maximum (highest pitch) on the final syllable of the whphrase (marking the end of the focus) to the final syllable of the utterance. Tone copying shows up when the focused constituent is not in final position and occurs also in declaratives and yes/no questions (Beyssade et al. 2004; Clech-Darbon et al. 1999; Delais-Roussarie et al. 2004; Doetjes et al. 2004).

which a speaker knows that both Anne and Sophie have booked a restaurant for Thursday evening, to have dinner with their respective families. In this context, the in-situ wh-phrase in (41) is licensed as choice function wh-in-situ, because it has been established in the context that there is a specific restaurant where the addressee made the reservation.

#### (41)Speaker addressing Sophie:

Anne m' a dit gu'elle a réservé au Pavillon pour jeudi Anne me has said that she has booked at the Pavillon for Thursday soir. Et toi. tu as réservé auel resto pour evening and you you have booked which restaurant for soir? ieudi

Thursday evening

'Anne told me she's booked the restaurant Pavillon for Thursday evening. And what/which restaurant did you book for Thursday evening?" (adapted from Glasbergen-Plas et al. 2021: 581, Fig. 3)

In the described context, the wh-in-situ question has a narrow focus on the word quel 'which', as resto 'restaurant' and the rest of the sentence are contextually salient. Thus, a context that licenses the use of the contextually bound choice function elicits questions with a narrow focus on the wh-word.

As this large final rise has been shown to be part of the characteristic tune for utterances with a narrow focus on the wh-word (Glasbergen-Plas et al. 2021), the fact that choice function wh-in-situ questions have a narrow focus on the wh-word suggests that these questions tend to be marked by a large sentence-final rise.<sup>23</sup> Consequently, for Type A speakers, who only have choice function wh-in-situ, French wh-in-situ questions typically display such a rise. Covert movement wh-in-situ questions may occur in various contexts, including out-of-the-blue cases (Glasbergen-Plas 2021; see Section 5.1), hence may also have other focus structures. Given that the contour used in the broad focus condition in Glasbergen-Plas et al. (2021) did not end in a large final rise, such a rise is not a typical feature of French wh-in-situ questions for Type B speakers. Thus, the 'Two speaker groups hypothesis' sheds new light on the longstanding debate concerning the alleged presence of a final rise. The results of Glasbergen-Plas et al. (2021) suggest that the authors who typically

<sup>23</sup> As one of the reviewers points out, one-to-one relations between syntactic structures and prosodic realisations are desired by syntacticians, but typically not found by phonologists. The study reported in Glasbergen-Plas et al. (2021) confirms this: the characteristic tunes used for the narrow focus condition and the broad focus condition were each used in about 70 % of the utterances in the corresponding condition. In line with the reviewer's remark, we suspect that the claims made in the syntactic literature about the necessity of a prosodic rise in wh-in-situ questions are too strong.

observed a falling contour (e.g. Mathieu 2002) investigated questions with broad focus.

# 5.3 Differences among speakers: intervention effects

If a group of French speakers who have both the choice function and covert movement strategies (Type B speakers) are to interpret wh-in-situ questions, the question arises what the predictions are for this group regarding (apparent) intervention effects. The short answer to this is that the predictions for both speaker groups are the same. Type B speakers are not expected to accept a wider range of sentences containing interveners, since an intervener blocks covert movement (cf. (3) in Section 1). The only difference between the two speaker groups regarding interveners is that true intervention effects (i.e. not apparent intervention effects) can only arise for Type B speakers, because they have the covert movement option.

Nonetheless, both Type A and Type B speakers may in some cases accept a wh-insitu question without an intervener yet fail to accept it when an intervener is added. The argument for why this is so runs parallel to the argument in Section 4.2 about differences between interveners. We argued in Section 4.2 that, when confronted with a sentence without context, it is easier to envisage a context that makes a choice function recoverable for certain interveners, such as plusieurs N 'several N', than for others, such as ne que 'only' in (33), repeated here as (42).

(42)parlé qu' à Suzanne dans quelles circonstances? #Jean (n') a Jean NE has spoken only to Suzanne in what circumstances 'In what circumstances did Jean only speak to Suzanne?'

We furthermore suggest that it is often more difficult to envisage a context that makes the choice function recoverable for a sentence with an intervener than for one without an intervener. For (42) to be interpretable via a choice function, the context must establish a referent for particular circumstances under which Jean only spoke to Suzanne. In contrast, a choice function in the same sentence without the intervener (43) would be interpretable if the context makes salient particular circumstances under which Jean spoke to Suzanne. We suggest that the latter type of context is easier to envisage than the former, because *ne que* 'only' imposes quite a specific restriction on the context (as in the comparison between *plusieurs N* 'several *N*' and *ne que* 'only' in Section 4.2).

(43)Iean parlé Suzanne dans quelles circonstances? circumstances Jean has spoken to Suzanne in what 'In what circumstances did Jean only speak to Suzanne?'

We, therefore, predict that both types of speakers may accept a wh-in-situ question without an intervener. The acceptance goes down when an intervener is added, because speakers may no longer be able to envisage the necessary context.

# 5.4 Differences among speakers: summary

The properties of wh-in-situ questions for the two groups of speakers are summarised in Table 1. The proposal sheds light on a number of longstanding debates between authors.

The first concerns the aforementioned 'extra-strong-presupposition'. Under our analysis, authors who argue for the presence of the presupposition (e.g. Boeckx 1999; Chang 1997; Cheng and Rooryck 2000) rely on judgements of Type A speakers while authors who argue against such a presupposition (e.g. Adli 2006; Mathieu 2004; Shlonsky 2012) rely on those of Type B speakers. In the absence of an intervener, Type A speakers still need an appropriate context to make the choice function recoverable, while Type B speakers do not, because they can rely on covert movement.

The second debate concerns the large sentence-final rise (Cheng and Rooryck 2000; Déprez et al. 2013). Under our analysis, choice function wh-in-situ questions always have a narrow focus on the wh-word. A large final rise is a prosodic correlate of this focus structure (Glasbergen-Plas et al. 2021). In other words, for Type A speakers, wh-in-situ questions always display a large final rise, while for Type B speakers, various prosodic tunes are possible depending on the context, including a falling contour.

Lastly, we have shown that it is easier for speakers to contextually establish a unique referent for the wh-phrase for certain interveners than for others. This explains the observed variation in judgements between different interveners (e.g. Baunaz 2005, 2011). We also illustrated how different speakers may envisage and

		Type A speakers	Type <b>B</b> speakers	
Licensing mechanism		Choice function	Choice function	Covert Movement
Extra-strong-presupposition		Yes	Yes	No
Intervener – f recoverable		OK	OK	No
Intervener – f not recoverable		No	No	No
Intonation	Large S-final rise	Yes	Yes	No
	No large S-final rise	No	No	Yes

**Table 1:** Overview of the properties of *wh*-in-situ for the two hypothesised groups of speakers.

interpret contexts differently. This sheds light on the observed speaker variation regarding intervention effects (e.g. Adli 2006; Mathieu 2002). It is important to realise that interveners do not differentiate between the two groups of speakers: both groups need to interpret a wh-in-situ question containing an intervener via a choice function, as an intervener blocks covert moment.

There seems to be a tendency for older literature to present data that, under the current analysis, represent a Type A grammar (only a choice function) (e.g. Boeckx et al. 2001; Chang 1997; Cheng and Rooryck 2000), whereas later literature is more often consistent with a Type B grammar (also covert movement) (e.g. Baunaz 2011; Reinhardt 2019). Even though this is far from a clear correlation across the relevant publications, we suggest that the picture is consistent with the idea that the language is changing: the grammar of Type A speakers may reflect an earlier variety of colloquial French. This variety would be a wh-fronting language that only allows for contextually restricted wh-in-situ. Possibly, contextually restricted whin-situ was gradually used more freely, for instance in the case of 'semi-salience' (see Section 4.3). Thus, a Type B grammar would emerge, in which covert movement is permitted to resolve in-situ wh-phrases, leading to felicitous use of out-ofthe-blue wh-in-situ questions by a subset of speakers. This scenario is in accordance with observations by Larrivée (2019) and the corpus data presented in Baunaz and Bonan (2023).

# 6 Conclusion and discussion

This paper has put forward a novel account of French wh-in-situ questions. We propose that the grammar of French contains a contextually bound choice function. The choice function allows a wh-phrase to be interpreted in-situ. However, the choice function's value must be uniquely specified by the context to be recoverable for the interlocutor. We suggest that this is the origin of what has so far been described as the 'extra-strong presupposition' of French wh-in-situ questions (Chang 1997; Cheng and Rooryck 2000; Coveney 1989a, b). The proposed account explains the observation that a French wh-in-situ question containing an intervener may become acceptable when placed in a certain type of context (Baunaz 2005, 2011, 2016; Beyssade 2006; Engdahl 2006; Starke 2001). We further hypothesise that there are two groups of French speakers ('Two speaker groups hypothesis'). While Type A speakers can only interpret wh-in-situ questions via the choice function, Type B speakers can additionally interpret wh-in-situ via covert

PROPERTY	Account
Absence of intervention effects in certain contexts	The referent for the <i>wh</i> -phrase must be recoverable in the context to interpret the choice function.
Extra-strong presupposition	The referent for the $\it wh$ -phrase has already been established in the context.
Intervention effects: differences between	Interveners differ in how easy it is for speakers to construe the
interveners	necessary context.
Intervention effects: differences among speakers	Speakers may vary in their ability to envisage and interpret contexts ('semi-salience').
Speaker variation regarding the extra-strong presupposition	Two speaker groups hypothesis: covert movement is only available for a subset of speakers.
Speaker variation regarding the presence of a large final rise	Two speaker groups hypothesis + experimental results of Glasbergen-Plas et al. (2021)

movement. Table 2 provides an overview of the properties of French wh-in-situ questions explained by the current account.

# 6.1 Comparison with Baunaz (2005, 2011, 2016), building on Starke (2001)

We now compare our account to the proposal by Baunaz (2005, 2011, 2016), which builds on Starke (2001) (henceforth: B/S). B/S propose a hierarchy of nanosyntactic features for both in-situ wh-phrases and interveners. As Baunaz (2016: 164) states, a wh-phrase 'that has something more than a potential intervener is free to move, whereas a [wh-phrase] that has either something less, or that has a similar feature composition as a potential intervener, will be blocked'. Specifically, Baunaz distinguishes between 'specific', 'partitive' and 'non-presuppositional' wh-phrases and interveners, where 'specific' has 'something more' than 'partitive' and 'partitive' has 'something more' than 'non-presuppositional'. Therefore, a 'partitive' wh-phrase would be able to cross a 'non-presuppositional' intervener, but not a 'specific' one.

The example in (44) illustrates for Baunaz the extraction of a partitive *wh*-phrase.

<sup>24</sup> The answer to a question with a 'specific' wh-phrase refers to a particular individual that the speaker already has in mind - the speaker merely asks for the identity of this individual. A wh-phrase with the feature 'partitive' is felicitous in a context that mentions the existence of a pre-established set. The term 'non-presuppositional' refers to the absence of either a 'specific' or 'partitive' feature.

Claire is a regular at Rainbow gym. She goes there 3 times a week. As it is usually the case in these infrastructures, she has a coach. Her coach usually prepares a plan for the day, i.e., she needs to use all the machines listed. That day Claire is a bit tired and she practices slower than usual. At the end of the session, she goes to the coach and tells him that she could not use all the machines. The coach, who wanted to prepare the next session is a bit angry. He asks:

```
Bon, t' as <u>pas</u> utilisé quelle(s) machine(s)? Well you have not used which(PL) machine(PL) 'Well, which machine(s) didn't you use?' (adapted from Baunaz 2016: 154–155, (35), underlining ours)
```

In the original example, the *wh*-phrase is written as singular (*quelle machine*), and Baunaz assumes that this is a partitive *wh*-phrase: there is a pre-established set of machines (the ones that Claire did not use) and the singular *wh*-phrase picks out one of these machines. Partitive readings are not expected to be possible under the current proposal: as one can conceive of different subsets from the mentioned set, the reference of the *wh*-phrase is not determined, and thus not recoverable from the context.

However, as the plural markers are not pronounced, the question in (44) does not specify whether *quelle(s) machine(s)* is a singular or a plural form. In the given context, it is likely that the coach wants to know which machine(s) Claire did not use, as this is the information they need for preparing the program for the next session. If Claire did not use the rowing machine, this machine might for example be referred to as 'the rowing machine', 'that big one' or 'the last one in the circle'. On the other hand, if Claire did not use the rowing machine nor the cycling bike, these two form a plural individual that can for instance be referred to as 'the rowing machine and the cycling bike', 'the last two in the circle' or 'the ones she dislikes most'. Whichever description is used in the answer to the *wh*-question, any congruent answer refers to the referent that is identified by the context, the 'one or more machines that Claire did not use'. Hence, the referent is uniquely identifiable, and the question might be paraphrased by the identity question in (45), as in the case of the examples with specific *wh*-phrases (in Baunaz terms) discussed in Section 2.

(45) The machine or machines that you did not use – which is it or which are they?

We conclude that (44) is not a counterexample to our proposal and suggest that there is no need to assume the category 'Partitive' in Baunaz's hierarchy.

The current proposal has a number of important advantages over B/S's account. First, it accounts for more data. In contrast to B/S, our account explains the observed

variation in judgements among sentences containing the same intervener. Likewise, in contrast to B/S, it explains why a subset of speakers (Type A speakers in our terms) only accept wh-in-situ in a particular type of context, regardless of whether the question contains an intervener or not. Second, the current account explains the contextual effects found for wh-in-situ without assuming the presence of a [+specific] feature. The observed effects are accounted for by a felicity condition on the interpretation of the choice function, which needs to be recoverable for the interlocutor. Since the mechanism of a choice function is directly related to in-situness, the semantic effects are expected to occur for in-situ wh-phrases only. In B/S'z account, it is not clear why in-situ but not fronted wh-phrases exhibit the proposed feature hierarchy. Cross-linguistically, specificity correlates with movement. Both in Turkish (Enç 1991) and in Germanic languages (De Hoop 1992, among others), specific noun phrases typically undergo scrambling or object shift, while non-specific noun phrases do not. The presence of a feature [+specific], as proposed by B/S, is therefore not expected to correlate with in-situness. Finally, given the 'Intervention effects avoidance generalisation' as stated in (16), we expect that the referent for the whphrase is recoverable in the context. Within the proposal of B/S, this descriptive generalisation should be an effect of the specificity of the wh-phrase, but it is not clear why specificity would have such an effect, given that specific noun phrases can be used in sentences with different types of focus-ground articulations.

# 6.2 Cross-linguistic look-ahead

In this last section, we turn to the cross-linguistic picture of wh-in-situ in wh-fronting languages. In what languages would choice function single wh-in-situ questions be expected to emerge? We expect that a wh-fronting language that has multiple wh-questions (such as English) can develop contextually restricted single wh-in-situ. Multiple wh-questions, with one fronted and one in-situ wh-expression, have two possible readings: single pair and pair-list. According to Dayal (2002), both choice functions and covert wh-movement are needed to derive these two readings. In multiple wh-questions, the choice function variable is unselectively bound by the existential quantifier introduced by the fronted wh-phrase. In the absence of a fronted wh-phrase, however, there is no binder within the sentence. We assume that in that case, contextual binding may become an option resulting in contextually restricted wh-in-situ.

This expectation seems to be borne out. Several languages with multiple wh-questions have been reported to allow a restricted use of wh-in-situ, which is only acceptable in a specific set of contexts (Biezma 2018; Bobaljik and Wurmbrand 2015; Bolinger 1978; Ginzburg and Sag 2000; Jiménez 1997; Poschmann 2015; Roussou et al. 2014). These languages include English, German, Spanish, (Brazilian) Portuguese and Modern Greek.

We suggest that the generalisation presented in Section 2 for French also goes a long way towards analysing the contextual restrictions of wh-in-situ questions in English and German, which are described in detail in the literature. In (46) and (48), the referent for the wh-phrase is already established is the context, as illustrated by the identity questions in (47) and (49), respectively.

- (46)A: I'm going to send the sourdough bread to the Southern Bakery, and the croissants to Barringers.
  - B: I see, and the bagels you're going to send WHERE? (Ginzburg and Sag 2000: 280, (65))
- (47)The place where you're going to send the bagels – what is it?
- (48)*Major*, you want this stuff WHERE? (Bobaljik and Wurmbrand 2015: 14, (2b))
- The place where you want the stuff what is it? (49)

As with contextually restricted wh-in-situ in French, the acceptability of examples such as (46) and (48) seems to vary, with certain speakers being able to accept wh-in-situ more easily than others. Hence, the choice function account proposed for French may also be a promising direction of research for wh-in-situ in languages such as English and German. If this contextually restricted wh-in-situ is gradually used more freely, wh-in-situ in such languages might stand at the beginning of a similar language change as the one we proposed for French in Section 5. Future research will have to show whether the choice function analysis developed in this paper can be extended to other languages and whether the further development towards a system allowing covert movement in single wh-questions can also be observed for other languages than French.

**Acknowledgements:** We thank Romane Pedro and Marie Pedro for helping us construct some of the examples and for contacting other speakers, who confirmed their judgements. We are also indebted to Benjamin Storme, Stéphane Térosier and Céline Zaepffel for fruitful discussion on the data in this paper. We further thank the audience of a talk at the University of Nantes, as well as the audiences of Sensus 2 and the MECORE Kickoff workshop, where earlier versions of the analysis were presented. Finally, we thank three anonymous Probus referees for their useful comments and their time.

**Author contributions:** All authors have accepted responsibility for the entire content of this manuscript and approved its submission.

**Research funding:** This research has been part of the project Understanding Questions (360-70-480), funded by the Dutch Research Council (NWO).

# References

- Adli, Aria. 2004. Y a-t-il des morphèmes intonatifs impliqués dans la syntaxe interrogative du français? Le cas du qu-in-situ. In Trudel Meisenburg & Maria Selig (eds.), Nouveaux Départs En Phonologie: Les Conceptions Sub-Et Suprasegmentales, 199–216. Tübingen: Gunter Narr.
- Adli, Aria. 2006. French wh-in-situ questions and syntactic optionality: Evidence from three data types. Zeitschrift für Sprachwissenschaft 25(2). 163-203.
- Aoun, Joseph, Norbert Hornstein & Dominique Sportiche. 1981. Some aspects of wide scope quantification. *Journal of Linauistic Research* 1(3), 69-95.
- Baunaz, Lena. 2005. The syntax and semantics of wh in-situ and existentials: The case of French. Leiden Papers in Linguistics 2(2). 1-27.
- Baunaz, Lena. 2011. The grammar of French quantification. Dordrecht: Springer Science & Business Media.
- Baunaz, Lena. 2016. French "quantifiers" in questions: Interface strategies. Acta Linguistica Hungarica 63(2), 125-168.
- Baunaz, Lena & Caterina Bonan. 2023. Activation levels. A fresh perspective on French wh in-situ. Isogloss 9(1), 1-43.
- Beck, Sigrid. 1996. Quantified structures as barriers for LF movement. Natural Language Semantics 4(1).
- Beyssade, Claire. 2006. La structure de l'information dans les questions: quelques remarques sur la diversité des formes interrogatives en français. Linx. Revue Des Linquistes De L'université Paris X Nanterre 55. 173-193.
- Beyssade, Claire, Elisabeth Delais-Roussarie, Jenny Doetjes, Jean-Marie Marandin & Annie Rialland. 2004. Prosody and information in French. In Francis Corblin & Henriëtte de Swart (eds.), Handbook of French semantics, 477-499. Stanford: CSLI Publications.
- Biezma, María. 2018. Givenness and the difference between wh-fronted and wh-in-situ questions in Spanish. In Janine Berns, Haike Jacobs & Dominique Nouveau (eds.), Romance languages and linguistic theory 13, 21-39. Amsterdam: John Benjamins.
- Bobaljik, Jonathan David & Susi Wurmbrand. 2015. Questions with declarative syntax tell us what about selection? In Ángel J. Gallego & Dennis Ott (eds.), 50 years later: Reflections on chomsky's aspects, 13-31. Cambridge, MA: MIT Working Papers in Linguistics.
- Boeckx, Cedric. 1999. Decomposing French questions. University of Pennsylvania Working Papers in Linguistics 6(1). 69-80.
- Boeckx, Cedric, Penka Stateva & Artur Stepanov. 2001. Optionality, presupposition, and WH-in situ in French. Amsterdam Studies in the Theory and History of Linguistic Science Series 4. 57–72.
- Bolinger, Dwight. 1978. Asking more than one thing at a time. In Henry Hiz (ed.), Questions, 107-150. Dordrecht: D. Reidel.
- Bošković, Željko. 1998. LF movement and the minimalist program. In Proceedings of NELS 1998, 43-57. Amherst: GLSA, University of massachusetts.
- Bošković, Željko. 2000. Sometimes in spec, CP, sometimes in situ. In Roger Martin, David Michaels, Juan Uriagereka & Samuel Jay Keyser (eds.), Step by step: Essays on minimalist syntax in honor of howard lasnik, 53-87. Cambridge, MA: MIT Press.

- Boucher, Paul. 2010. Wh-questions in French and English: Mapping syntax to information structure. In Carsten Breul & Edward Göbbel (eds.), Comparative and contrastive studies of information structure, 101-137. Amsterdam/Philadelphia: John Benjamins.
- Breheny, Richard. 2003. Exceptional-scope indefinites and domain restriction. In Matthias Weisgerber (ed.), Proceedings of Sinn Und Bedeutung 7, 38-52. Constance: University of Constance.
- Büring, Daniel. 2016. Intonation and meaning. Oxford: Oxford University Press.
- Chang, Lisa. 1997. Wh-in-situ phenomena in French. Vancouver: University of British Columbia MA thesis. Cheng, Lisa L.-S. & Johan Roorvck, 2000. Licensing wh-in-situ, Syntax 3(1), 1–19.
- Clech-Darbon, Anne, Georges Rebuschi & Annie Rialland. 1999. Are there cleft sentences in French. In Georges Rebuschi & Laurice Tuller (eds.), The grammar of focus, 83–118. Amsterdam: John Benjamins.
- Coveney, Aidan. 1989a. Pragmatic constraints on interrogatives in spoken French. York Papers in Linguistics
- Coveney, Aidan. 1989b. Variability in interrogation and negation in spoken French. Newcastle: Newcastle University Dissertation.
- Dayal, Veneeta. 2002. Single-pair versus multiple-pair answers: Wh-in-situ and scope. Linguistic Inquiry 33(3), 512-520,
- Dayal, Veneeta. 2016. Questions. Oxford: Oxford University Press.
- de Hoop, Helen. 1992. Case configuration and noun phrase interpretation. Groningen: University of Groningen Dissertation Published in 1996 by Garland.
- Deemter, Kees van. 1994. What's new? A semantic perspective on sentence accent. Journal of Semantics 11(1-2), 1-32,
- Delais-Roussarie, Elisabeth, Jenny Doetjes & Petra Sleeman. 2004. Dislocation. In Francis Corblin & Henriëtte de Swart (eds.), *Handbook of French semantics*, 501–528. Stanford: CSLI Publications.
- Déprez, Viviane, Kristen Syrett & Shigeto Kawahara. 2013. The interaction of syntax, prosody, and discourse in licensing French wh-in-situ questions. Lingua 124. 4–19.
- Di Cristo, Albert. 1998. Intonation in French. In Daniel Hirst & Albert Di Cristo (eds.), Intonation systems: A survey of twenty languages, 195–218. Cambridge: Cambridge University Press.
- Doetjes, Jenny, Georges Rebuschi & Annie Rialland. 2004. Cleft sentences. In Francis Corblin & Henriëtte de Swart (eds.), Handbook of French semantics, 529-552. Stanford: CSLI publications.
- Eckardt, Regine. 2007. Inherent focus on wh-phrases. In Estela Puig-Waldmueller (ed.), Sinn und Bedeutung 11, 209-228. Barcelona: Universitat Pompeu Fabra.
- Enç, Mürvet. 1991. The semantics of specificity. *Linguistic Inquiry* 22(1). 1–25.
- Engdahl, Elisabet. 2006. Information packaging in questions. In Olivier Bonami & Patricia Cabredo Hofherr (eds.), Empirical issues in formal syntax and semantics, 93–111. Paris: CSSP.
- Faure, Richard & Katerina Palasis. 2021. Exclusivity! Wh-fronting is not optional wh-movement in colloquial French. Natural Language and Linguistic Theory 39. 57-95.
- Ginzburg, Jonathan & Ivan A. Sag. 2000. Interrogative investigations. Stanford: CSLI Publications.
- Glasbergen-Plas, Aliza. 2021. Questions in context: The case of French wh-in-situ. Leiden: Leiden University Dissertation Utrecht: LOT.
- Glasbergen-Plas, Aliza, Stella Gryllia & Jenny Doetjes. 2021. The prosody of French wh-in-situ questions: Echo vs. non-echo. Journal of Linguistics 57(3). 569-603.
- Guerzoni, Elena. 2006. Intervention effects on NPIs and feature movement: Towards a unified account of intervention. Natural Language Semantics 14. 359-398.
- Hamblin, Charles Leonard. 1973. Questions in montague English. Foundations of Language 10. 41-53.
- Hamlaoui, Fatima. 2010. Anti-givenness, prosodic structure and "intervention effects". The Linguistic Review 27(3). 347-364.

- Hamlaoui, Fatima. 2011. On the role of phonology and discourse in Francilian French wh-questions. Journal of Linguistics 47(1). 129-162.
- Heim, Irene. 2000. Notes on interrogative semantics. Lecture notes for advanced semantics. MIT.
- Heim, Irene. 2008. Artikel und Definitheit. In Arnim von Stechow & Dieter Wunderlich (eds.), Semantik/ semantics, 487-535. Berlin/New York: De Gruyter Mouton.
- Ionin, Tania. 2006. This is definitely specific: Specificity and definiteness in article systems. Natural Language Semantics 14(2). 175-234.
- Jacobs, Joachim. 1984. Funktionale Satzperspektive und Illokutionssemantik. Linguistische Berichte 91. 25-58.
- Jacobs, Joachim. 1991. Implikaturen und 'alte Information' in w-Fragen. In Marga Reis & Inger Rosengren (eds.), Fragesätze Und Fragen, 201–221. Tübingen: Niemeyer.
- Jiménez, María Luisa. 1997. Semantics and pragmatics conditions on word order in Spanish. Washington DC: Georgetown University Dissertation.
- Karttunen, Lauri. 1977. Syntax and semantics of questions. Linguistics and Philosophy 1(1). 3-44.
- Kotek, Hadas. 2018. Composing questions. Cambridge, MA: MIT Dissertation Published in 2019 by MIT Press.
- Kratzer, Angelika. 1998. Scope or pseudoscope? Are there wide scope indefinites? In Susan Rothstein (ed.), Events and arammar. Dordrecht: Kluwer.
- Ladd, D. Robert. 2009. Intonational phonology. Cambridge: Cambridge University Press.
- Larrivée, Pierre. 2019. Historical pragmatics, explicit activation and wh in situ in French. In Ingo Feldhausen, Martin Elsig, Imme Kuchenbrandt & Mareike Neuhaus (eds.), Romance languages and linguistic theory 15. Selected papers from going romance 30, Frankfurt, 114–132. Amsterdam: Benjamins.
- Martin, Philippe. 1981. Pour une théorie de l'intonation. In Mario Rossi, Albert Di Cristo, Daniel Hirst, Philippe Martin, Y. Nishinuma & L' intonation (eds.), De L'acoustique À La Sémantique, 234–271. Paris:
- Mathieu, Éric. 1999. WH in situ and the intervention effect. UCL Working Papers in Linguistics 11. 441–472.
- Mathieu, Éric. 2002. The syntax of non-canonical quantification: A comparative study. London: University of London Dissertation.
- Mathieu, Éric. 2004. The mapping of form and interpretation: The case of optional wh-movement in French. Lingua 114(9). 1090-1132.
- Myers, Lindsy Lee. 2007. Wh-interrogatives in spoken French: A corpus-based analysis of their form and function. Austin: The University of Texas at Austin Dissertation.
- Obenauer, Hans-Georg. 1976. Etudes de syntaxe interrogative du français: quoi, combien et le complémenteur. Tübingen: Niemeyer.
- Oiry, Magda. 2011. A case of true optionality: Wh in situ patterns like long movement in French. Linquistic Analysis 37(1-2). 1-28.
- Poletto, Cecilia & Jean-Yves Pollock. 2015. Arguing for remnant movement in Romance. In Günther Grewendorf (ed.), Remnant movement, 135-178. Berlin: Mouton de Gruyter.
- Poschmann, Claudia. 2015. Echo-Fragen: vom Satztyp zur Fragebedeutung. Berlin: De Gruyter.
- Reich, Ingo. 2002. Question/answer congruence and the semantics of wh-phrases. Theoretical Linguistics 28(1). 73-94.
- Reinhardt, Janina. 2019. Regularity and variation in French direct interrogatives: An analysis of the morphosyntax and intonation of question forms in reality TV shows, audio books, and teaching materials. Konstanz: University of Konstanz Dissertation.
- Reinhart, Tanya. 1998. Wh-in-situ in the framework of the minimalist program. Natural Language Semantics 6(1). 29-56.

- Rochemont, Michael S. 1986. Focus in generative grammar (studies in generative linguistic analysis 4). Amsterdam/Philadelphia: John Benjamins Publishing Company.
- Rothstein, Susan, 2001, Predicates and their subjects, Dordrecht; Kluwer,
- Roussou, Anna, Christos Vlachos & Dimitris Papazachariou. 2014. In situ, ex situ and (non-)echo questions. In Nikolaos Lavidas, Thomaï Alexiou & Areti Maria Sougari (eds.), Major trends in theoretical and applied linguistics: Selected papers from the 20th international symposium on theoretical and applied linguistics (Istal 20), 475–494. London: Versita de Gruyter.
- Rowlett, Paul. 2007. The syntax of French. Cambridge: Cambridge University Press.
- Schwarzschild, Roger. 1999. GIVENness, AvoidF and other constraints on the placement of accent. Natural Language Semantics 7(2), 141-177.
- Selkirk, Elisabeth O. 1984. Phonology and syntax: The relationship between sound and structure. Cambridge, MA: MIT Press.
- Shlonsky, Ur. 2012. Notes on wh in situ in French. In Laura Brugé, Anna Cardinaletti, Giuliana Giusti, Nicola Munaro & Cecilia Poletto (eds.), Functional heads: The cartography of syntactic structures, 242-252. Oxford: Oxford University Press.
- Starke, Michal. 2001. Move reduces to merge: A theory of locality. Geneva: University of Geneva Dissertation. Tual, Lucas. 2017. Prosodic and syntactic properties of French interrogative sentences. In *Paper presented* at 43rd Incontro Di Grammatica Generativa, Pavia, Italy,
- Winter, Yoad. 1997. Choice functions and the scopal semantics of indefinites. Linguistics and Philosophy 20. 399-467.
- Wunderli, Peter. 1978. Das Forschungsprojekt Intonation im Französischen (Le projet de recherche sur l'intonation en français). Zeitschrift für Romanische Philologie Tübingen 94(1–2). 93–118.
- Wunderli, Peter. 1982. Die Intonation der Fragen vom Typ 'Tu penses à quoi?' In Otto Winkelmann & Maria Braisch (eds.), Festschrift für Johannes Hubschmid zum 65. Geburtstag, 169-181. Bern & München: Francke.
- Wunderli, Peter. 1983. L'intonation des phrases interrogatives du type 'Il est né en quelle année. Romanica Gandensia 20. 169-181.
- Wunderli, Peter & Petra Braselmann. 1980. L'intonation des phrases interrogatives: le type 'Tu vas où. Studii si Cercetări Linguistice 31(5). 649-660.
- Yanovich, Igor. 2005. Choice-functional series of indefinite pronouns and Hamblin semantics. In Effi Georgala & Jonathan Howell (eds.), *Proceedings of Salt 15, Ithaca*, 309–326. Cornell: Cornell Linguistics Publications.
- Zimmermann, Malte & Edgar Onea. 2011. Focus marking and focus interpretation. Lingua 121(11). 1651-1670.
- Zribi-Hertz, Anne. 1994. La syntaxe des clitiques nominatifs en français standard et en français avancé. Travaux de Linquistique et de Philologie XXXII. 131–147.
- Zubizarreta, María Luisa. 2003. Intervention effects in the French wh-in-situ construction: Syntax or interpretation? In Rafael Nuñez-Cedeño, Luis López & Richard Cameron (eds.), A romance perspective on language knowledge and use: Selected papers from the 31st linguistic symposium on romance languages, 359-379. Amsterdam/Philadelphia: John Benjamins.